Author's response to reviews

Title: Association between family environment and Attention Deficit Hyperactivity Disorder in children - mothers’ and teachers’ views

Authors:

Thiago de O Pires (thop100@hotmail.com)
Cosme MFP da Silva (cfpassos@gmail.com)
Simone G de Assis (sgassis@globo.com)

Version: 2 Date: 14 August 2013

Author's response to reviews:

The questions were answered in the letter.